Author’s response to reviews

Title: Back pain outcomes in primary care following a practice improvement intervention: a prospective cohort study

Authors:

Alan C Breen (imrci.abreen@aecc.ac.uk)
Eloise Carr (ECarr@bournemouth.ac.uk)
Jennifer E Langworthy (imrci.jlangworthy@aecc.ac.uk)
Clive Osmond (co@mrc.soton.ac.uk)
Louise Worswick (lworswick@bournemouth.ac.uk)

Version: 8 Date: 26 January 2011

Author’s response to reviews: see over
Comments on final revisions

These revisions are solely for punctuation and making the font in the first paragraph of the Discussion section conform to the rest of the paper. I have also altered the line settings of the Acknowledgements and References so that they start at the top of the pages.